

A Service of

ZBU

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

García-Morán, Eva; Kuehn, Zoë

Working Paper With strings attached: Grandparent-provided child care and female labor market outcomes

SOEPpapers on Multidisciplinary Panel Data Research, No. 610

Provided in Cooperation with: German Institute for Economic Research (DIW Berlin)

Suggested Citation: García-Morán, Eva; Kuehn, Zoë (2013) : With strings attached: Grandparentprovided child care and female labor market outcomes, SOEPpapers on Multidisciplinary Panel Data Research, No. 610, Deutsches Institut für Wirtschaftsforschung (DIW), Berlin

This Version is available at: https://hdl.handle.net/10419/88578

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



SOEPpapers

on Multidisciplinary Panel Data Research

SOEP - The German Socio-Economic Panel Study at DIW Berlin

610-2013

510 more statements of the second s

With Strings Attached: Grandparent-Provided Child Care and Female Labor Market Outcomes

Eva García-Morán and Zoë Kuehn



SOEPpapers on Multidisciplinary Panel Data Research at DIW Berlin

This series presents research findings based either directly on data from the German Socio-Economic Panel Study (SOEP) or using SOEP data as part of an internationally comparable data set (e.g. CNEF, ECHP, LIS, LWS, CHER/PACO). SOEP is a truly multidisciplinary household panel study covering a wide range of social and behavioral sciences: economics, sociology, psychology, survey methodology, econometrics and applied statistics, educational science, political science, public health, behavioral genetics, demography, geography, and sport science.

The decision to publish a submission in SOEPpapers is made by a board of editors chosen by the DIW Berlin to represent the wide range of disciplines covered by SOEP. There is no external referee process and papers are either accepted or rejected without revision. Papers appear in this series as works in progress and may also appear elsewhere. They often represent preliminary studies and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be requested from the author directly.

Any opinions expressed in this series are those of the author(s) and not those of DIW Berlin. Research disseminated by DIW Berlin may include views on public policy issues, but the institute itself takes no institutional policy positions.

The SOEPpapers are available at http://www.diw.de/soeppapers

Editors:

Jürgen **Schupp** (Sociology) Gert G. **Wagner** (Social Sciences, Vice Dean DIW Graduate Center)

Conchita **D'Ambrosio** (Public Economics) Denis **Gerstorf** (Psychology, DIW Research Director) Elke **Holst** (Gender Studies, DIW Research Director) Frauke **Kreuter** (Survey Methodology, DIW Research Professor) Martin **Kroh** (Political Science and Survey Methodology) Frieder R. **Lang** (Psychology, DIW Research Professor) Henning **Lohmann** (Sociology, DIW Research Professor) Jörg-Peter **Schräpler** (Survey Methodology, DIW Research Professor) Thomas **Siedler** (Empirical Economics) C. Katharina **Spieß** (Empirical Economics and Educational Science)

ISSN: 1864-6689 (online)

German Socio-Economic Panel Study (SOEP) DIW Berlin Mohrenstrasse 58 10117 Berlin, Germany

Contact: Uta Rahmann | soeppapers@diw.de

With Strings Attached: Grandparent-Provided Child Care and Female Labor Market Outcomes^{*}

Eva García-Morán[†] Zoë Kuehn[‡]

November 2013

Abstract

Grandparents are regular providers of free child care. Similar to other forms of child care, availability of grandparent-provided child care affects fertility and labor force participation of women positively. However, grandparent-provided child care requires residing close to parents or in-laws which may imply costly spatial restrictions. We find that mothers residing close to parents or in-laws have lower wages and that the probability of having to commute increases if relatives provide child care. We build a model of residence choice, fertility, and female labor force participation that can account for the observed relationships. We simulate our model to analyze how women's decisions would change if the availability of grandparent-provided child care or family policies were altered. We find that if child care subsidies were raised to the Swedish level, fertility and mothers' labor force participation would increase, while mobility would remain unchanged. The absence of grandparents, on the other hand, would increase mobility, while it would only have limited negative effects on aggregate fertility and labor force participation.

JEL classification: J13, J61, H42, R23

Keywords: grandparent-provided child care, fertility, labor force participation, spatial restrictions, regional labor markets

†
eva.garcia@unibocconi.it \cdot IGIER \cdot Bocconi
 University \cdot Via Roentgen 1 \cdot 20136, Milan
 \cdot Italy.

^{*}Earlier versions of this paper have been awarded UWIN Unicredit's Prize for Best Paper in Gender Economics 2012 and the Etta Chiuri Prize 2012. We would like to thank all participants of the Workshop on Gender Equality at Bocconi University for their interesting comments and observations, especially Barbara Petrongolo, Alberto Alesina. We are grateful to Vincenzo Galasso for his useful comments. We would also like to thank two anonymous referees for their useful comments. Zoë Kuehn gratefully acknowledges financial aid by FEDEA (Fundación de Estudios de Economía Aplicada) in the context of the project "Evaluación de Políticas Públicas (ECO2011-30323-C03-01) of the Spanish Ministry of Science and Research.

 $^{^{\}ddagger}$ zoe.
kuehn@uam.es \cdot Universidad Autónoma de Madrid
 \cdot Departamento de Análisis Económico: Teoría Económica
 \cdot Campus de Cantoblanco \cdot 28049
 Madrid \cdot Spain.

1 Introduction

Grandparents are an important source of child care. According to data from the 2^{nd} wave of the Survey of Health, Ageing and Retirement in Europe (SHARE), between 18% (Denmark) and 49% (Italy) of grandparents take care of their grandchildren age six or younger on a daily or weekly basis. More than 30% of grandparents in the Netherlands, Belgium, and Switzerland provide weekly care, whereas in Italy, Greece, and Poland almost 30% look after their grandchildren each day (see Figure A-1 of the Appendix A).¹ The availability of child care and especially cheap or even costless child care has important effects on fertility and mothers' labor force participation.² For Italy, Del Boca [2002] shows that both the availability of child care and the possibility of part time work increase labor force participation and fertility. Blau and Robins [1989] establish a similar pattern for the US. Within the context of already high female labor force participation rates in Sweden, Mörck et al [2009] is one of the few papers that focuses exclusively on the positive effect of lower child care costs on fertility. In a literature summary, Del Boca and Vuri [2007] point out that most studies find that high child care costs deter female labor supply, while availability of child care is found to have positive effects on mothers' labor force participation. These findings suggest that the main barrier that mothers face at the time of working is to obtain affordable child care.³

Free grandparent-provided child care seems to be the perfect solution for working mothers. However, in order to enjoy grandparent-provided child care on a regular basis, residence choices of adult children and elderly parents have to coincide. Data from the 2^{nd} wave of SHARE show that the frequency of grandparent-provided child care is clearly related to the geographical distance between parents and their adult children. Figure 1.1 displays the frequency of grandparent-provided child care, together with the geographical distance between grandparents and their grandchildren (age six and younger) for Germany, Denmark, Italy, and Spain. Across the four selected countries the frequency of care varies strongly, but common to all countries those who provide care more frequently tend to live close by.⁴

¹In the US, 22.7% of children under 5 are regularly cared for by their grandparents (Overturf Johnson [2005]).

²While female labor force participation has increased tremendously over the past decades, mothers are still participating significantly less. Across OECD countries, the average difference in labor force participation rates for women and mothers (of children age 3 or younger) is around 10 (20) percentage points. Scandinavian countries are an exception where labor force participation rates of mothers are equal to or even higher compared to those of women in general, OECD [2008].

³For instance, for a US family living below the poverty line, child care costs amount to 30% of income (US Census Bureau [2011]).

⁴The same pattern can be observed across the rest of the countries included in SHARE; see Figures A-3-A-2 of the Appendix A.



Figure 1.1: Frequency of care for and distance to grandchild, ≤ 6

Data: Survey of Health, Ageing and Retirement in Europe (SHARE), 2nd wave.

Grandparent-provided child care – similar to other types of child care – may have positive effects on fertility and mothers' labor force participation. However, – different from other types of child care – it imposes spatial restrictions which may affect female labor market outcomes negatively. Looking at German data we find that women residing close to parents or in-laws have lower hourly and monthly wages. Furthermore we find that mothers are more likely to incur in commutes if their children are regularly cared for by relatives. Similar to other studies, we also find that women residing close to parents or in-laws are more likely to have children, and that as mothers they are more likely to hold a regular full or part time job. Given strong interdependencies of decisions regarding residence, fertility, child care arrangements, and female labor force participation, our empirical analysis faces problems of endogeneity and reversed causality. While our findings can thus not be interpreted as causal, the observed relationships provide the motivation for our theoretical model.

We build a model of residence choice, fertility decisions, and female labor force participation. In our model there are two regions. Both regions are characterized by the same distribution of wage offers. Women who decide to work need to purchase child care. In one of the two regions free grandparent-provided child care is available. A woman decides where to reside upon observing her potential wage-were she to work- and her spouse's income in each region. Both her wage and her spouse's income are independent across regions. So when deciding where to reside women might accept lower wages in turn for free grandparent-provided child care. This mechanism generates the observed relationship between lower wages and living close to parents or in-laws. Given that women are more likely to have children and to work if child care is free, our model also replicates the positive relationships between fertility, labor force participation, and living close to parents or in-laws. In our model, the simultaneous interaction of child care costs with two important determinants of family migration decisions – husbands' and wives' wages – generates the observed relationships.⁵ We then simulate the model to analyze how women's decisions would change if the availability of grandparent-provided child care or family policies were altered. We find that if child care subsidies were raised to the Swedish level, fertility and mothers' labor force participation would increase, while mobility would remain unaltered. The absence of grandparents, on the other hand, would increase mobility, while it would only have limited negative effects on aggregate fertility and labor force participation.

To the best of our knowledge the current paper is the first one that explicitly incorporates spatial restrictions imposed by grandparent-provided child care into a model of fertility and labor force participation decisions. There is a strand of the literature analysing fertility and labor force participation in general equilibrium models as for instance Attanasio et al [2010], Erosa et al [2010], García-Morán [2010], Greenwood et al [2000], or Guner and Knowles [2009]. Closely related is Cardia and Ng [2003] who – different from the current paper – explicitly incorporate grandparents' decisions into a general equilibrium model of grandparent-provided child care. The authors suggest that subsidizing grandparents' time is the most effective child care policy in terms of output and capital accumulation. However, they do not consider the spatial restrictions and potential costs in terms of labor market outcomes implied by grandparent-provided child care. Also related to this paper are the works by Bick [2010] and Mendez [2008]. Within a life cycle model, the former analyzes data for Germany and concludes that informal child care (by relatives) plays an important role given that mothers' labor force participation exceeds child care enrollment for children up to 2 years. However, different from the current paper the author does not model relative-provided child care, nor the spatial restrictions that it imposes. The paper by Mendez [2008] attempts to account for differences in geographical mobility and female labor force participation across European countries. His model of residence choice, fertility, and female labor force participation is similar to ours, but the author does not provide any evidence for individual costs associated with living close to parents or in-laws.

⁵Mincer [1978] finds husbands' wage offers to be determinant for family migration decisions, but to a lesser extent if the wife is working and if her work is permanent and well paid. Hence, husbands' and wives' wages jointly determine migration decisions.

Thus, our paper is also - to the best of our knowledge - the first one to document possible costs related to the geographical proximity between parents and adult children. The existing literature, on the contrary, has highlighted its positive aspects. Studying fertility intentions rather than outcomes, Raymo et al [2010] find that Italian and Japanese women living close to their parents have higher fertility intentions. Holdswoth and Dale [2009] study labor force participation decisions of mothers in Spain and Britain. They estimate that the probability of being in employment is 1.24 times higher for Spanish women whose parents live in the same town. For the US, Compton and Pollak [2011] find that married women with small children living close to mothers and mothers-in-law have a 10 percentage point higher probability to be in employment. The studies by Dimova and Wolff [2011] and Zamarro [2011] use SHARE data and estimate simultaneous equation models of labor supply and grandparent-provided child care. Zamarro [2011] only finds a positive effect of grandparent-provided child care on mothers' labor force participation for Greece and the Netherlands. For ten European countries Dimova and Wolff [2011], who also include financial transfers into their model, find a positive effect of grandparent-provided child care on the extensive margin of female labor force participation but no effect along the intensive margin. Arpino et al [2010] and Posadas and Vidal-Fernández [2012] for Italy and the US respectively, find that grandparent-provided child care – instrumented by grandparents being alive – increases in particular labor force participation of low educated mothers of young children.

Our paper is also related to a broader literature on intergenerational time transfers. Most of this literature focuses on time transfers from children to elderly parents. One interesting paper that also incorporates residence choices is Konrad et al [2002]. The authors develop a game theoretical model of strategic choice of residence among siblings who try to avoid having to take care of elderly parents. Looking at German data, they find support for their model's predictions of older siblings locating further away from their parents than younger siblings. With a similar approach in mind, Stern [1995] estimates care choices of elderly parents together with location decisions of children. His work is closely related to the current paper as he also takes into account how the child's location decision affects his or her work decision. Alesina and Giuliano [2010] argue that the extent of intergenerational time transfers within the family – care of children or the elderly – is strongly determined by the value of family ties in a society. The authors find that across countries a higher value of family ties is associated with lower geographical mobility, higher fertility, but also with more traditional gender roles and thus lower female labor force participation. In the current paper – on the contrary – we find that within a country grandparent-provided child care is related to higher female labor force participation.

The remainder of this paper is organized as follows: the next section presents our empirical analysis, Section 3 presents the model, Section 4 describes our calibration strategy, Section 5 presents the results of the model, and in Section 6 we discuss the model's mechanisms in detail. In Section 7 we perform two counterfactual experiments and Section 8 concludes.

2 Empirical Analysis

2.1 Data

For our empirical analysis, we consider data from the German Socio-Economic Panel (SOEP), an annual household survey. The SOEP provides extensive information on individuals' labor market participation, marital and family status, wages, education, etc.⁶ It also includes variables of particular interest for our analysis: child care provided by relatives and geographical distance to parents. In four waves (1991, 1996, 2001, and 2006) survey participants were asked to categorize their parents' relative residence as in: i) the same house, ii) the same neighborhood, iii) the same town, iv) another town but within one hour by car, v) further away, or vi) in a foreign country. We construct a dummy variable "parents or in-laws close" that takes on value one for individuals whose mother, father, or in-law lives in the same neighborhood or town. For individuals who live in the same house as their parents or in-laws we construct a different dummy variable "parents or in-laws in same house." This particular form of co-residence mostly arises because young individuals still live at home or due to the need for intensive care of parents, and in most cases it represents a temporary living arrangement. Only in two of the four waves considered (2001 and 2006) did the survey also include questions about relative-provided child care. We construct a dummy variable "child care by relatives" that takes on value one for all mothers with children age six or younger if relatives regularly take care of this child and a dummy variable "child care non-relatives" that takes on value one if the child age six or younger attends a nursery or is being cared for by others than relatives. We report these variables for two different age groups, for children up to the age of three and those between the ages of three and six. In Germany, for children younger than three very few spots in public or publicly subsidized nurseries are available, whereas for children between the ages of 3 and 6, those spots are almost guaranteed.⁷

⁶For more details on the SOEP, see SOEP [2005] or Wagner et a [2007].

⁷In 2011 only 25.2% of children younger than three attended some form of private day care (85%) or public or publicly subsidized nurseries (15%), compared to 93.4% of children between three and six (less than 1% of them in private day care) (Statistische Ämter des Bundes und der Länder [2011]). However,

Given pronounced differences in mothers' labor force participation rates between East and West Germany, we introduce dummy variables to distinguish between individuals living in East and West Germany.⁸ To account for possible cultural differences, we also distinguish between individuals of German and other nationalities. We define three levels of education following the International Standard Classification of Education (ISCED 1997) designed by the UNESCO[1997]: (i) primary education (ISCED levels 0 and 1), (ii) secondary education (ISCED levels 2, 3, and 4), (iii) tertiary education (ISCED levels 5 and 6).⁹ Town sizes are grouped into small communities (up to 20.000 inhabitants), medium-sized communities (20.000-100.000 inhabitants), and large communities (more than 100.000 inhabitants). We also construct dummy variables for each federal state.

Sample For our analysis we pool the data from the available four waves. We only consider women age 25 to 50.¹⁰ We exclude those born outside of Germany, because for these individuals both key variables of our analysis, (i) availability of child care by relatives and (ii) residence relative to parents, might be determined by very different aspects. Our sample consists of 10,732 women and 8,129 mothers.¹¹

Descriptive Statistics Table A.1 of the Appendix A provides descriptive statistics for our sample. Individuals are on average between 37 and 39 years old. Approximately 76% of women between 25 and 50 have children and 10-13% are mothers of small children (age 0-2) and 14-19% have children age 3-6. Around 36% of women have a regular fulltime

most of these nurseries (for over 60% of children) provide only part-time arrangements, see European Commission [2009]. Compulsory schooling for German children starts at age 6-7, reducing the need for child care drastically. 'Regular child care by relatives' includes child care by any relative but it is mostly grandparents who provide child care. Similar spatial restrictions would apply for child care by other relatives.

⁸Labor force participation rates of East German mothers of small children (0-3 years) have traditionally been very high and even today they continue to be around 15 percentage points higher than rates for West German mothers (Bundesministerium für Familie, Senioren, Frauen und Jugend [2005].) In our sample, labor force participation rates of mothers are 67% and 48% in East and West Germany respectively.

⁹For our regression analysis we only differentiate between individuals with and without tertiary education.

¹⁰We exclude individuals living in East Germany from the 1991 wave, because for this wave information for most labor market variables (participation, wages) are missing for East Germans.

¹¹We also exclude individuals who report to have worked regular full-or part time jobs but who also report to have worked fewer than twenty hours a month as well as those who report to work regular full-or part time jobs but do not report their wage income or firm tenure or report zero or negative values for any of the two variables.

job and 24% hold a regular part time job. Mothers are more prone to hold part time jobs (29%) compared to fulltime jobs (25%).¹² Around 42-44% of women and mothers live in the same neighborhood or town as their parents or in-laws and 43-44% live at least one hour away. Approximately 13-14% of individuals live in the same house or household with parents or in-laws. Only 14% of mothers of children younger than age three, but 84% of mothers of children between the ages of three and six use nursery care, sitters or other types of paid child care. Around one third has their children cared for by relatives on a regular basis.¹³ Hourly wages of women and mothers are around $12 \in .^{14}$ Around 50% of individuals commute to their place of work.

Proximity to Grandparents and Relative-Provided Child Care Child care by relatives is not only likely to influence mothers' labor force participation decision, but at the same time its use might be determined by mothers' decision to work. This reversed causality introduces a potential bias into a direct measure of grandparent-provided child care. Using geographical proximity to grandparents as an indirect measure of grandparentprovided child care solves this problem. However, as residence choices might not be independent of mothers' labor force participation decisions, a caveat remains. Descriptives statistics show that geographical proximity to grandparents and child care provided by relatives are very much related (see Table A.2 of the Appendix A). The clear relationship displayed in Figure 1.1 regarding child care by relatives and proximity to grandparents hence also holds for our data set. We thus feel confident to use geographical proximity as an indirect measure of grandparent-provided child care. Furthermore, geographical proximity reflects more than just currently provided child care by grandparents. It might also reflect child care provided by "potential" grandparents if living close to parents or in-laws affects fertility decisions. Thus, geographical proximity proves particularly useful to test effects on fertility. In addition, geographical proximity might also reflect "child care provided in the past" if individuals continue to live close to parents or in-laws after children have grown beyond the child care age. In this case the indirect measure can also

¹²These figures correspond to 1991-2006 and are thus lower compared to more recent data. While in 1996, 75.7% of women and 58.5% of mothers age 25 to 54 participated in the German labor market, by 2004 these rates had increased to 77.0% and 63.1% respectively. This increase was almost exclusively due to an increase in labor force participation rates of mothers in West Germany from 54.4% in 1996 to 61.6% in 2004 (Bundesministerium für Familie, Senioren, Frauen und Jugend [2005]). Regarding labor force participation of mothers with children under 6, those increased from 51.4% in 1991 (OECD [2001]) to 66.3% by 2009 (OECD [2008]).

 $^{^{13}}$ Those two options are not exclusive. For children ages (0-2) and (3-6), 36% and 35% of mothers who use private care also have relatives looking after their children on a regular basis, and 15% and 86% respectively of those who use relative-provided care also use private care for their children.

¹⁴When pooling the sample we only adjust wages for the change from Deutschmark to Euro. We do not adjust for wage growth, given stagnant real hourly net wages in Germany between 2001 and 2006 (see Figure 1 in DIW [2009]).

be used to test long-lasting effects of on wages.

However, geographical proximity is only a necessary condition for grandparent-provided child care. Other aspects such as employment status, age, or health status of grandparents may determine if grandparents actually provide care for their grandchildren. Mothers of around 20% of women in our sample also participated in the SOEP during the four waves considered. The number of observations for these mother-daughter pairs is extremely limited and thus not suited for regression analysis. However, considering descriptive statistics we observe that grandmothers who live close to their daughters who in turn report to use relative-provided child care are younger, and they are also less likely to work (see Table A.3 of the Appendix A). Surprisingly, these grandmothers also report worse health conditions. This might be explained by the fact that grandmothers' status of health does not only determine if they provide child care, but caring for grandchildren may have a negative effect on grandmothers' health.¹⁵

2.2 Benefits of Grandparent-Provided Child Care

In line with findings in the literature discussed before, we find geographical proximity to ("potential") grandparents to affect fertility and labor force participation of mothers positively. Women living in the same neighborhood or town as their parents or in-laws have a 4 percentage points higher probability to have children (see Table A.4 of the Appendix A). We find this positive effect to be particularly strong for women with university education.¹⁶ Regarding participation, we find that mothers residing close to their parents or in-laws have a 3 percentage points higher probability to hold a regular part-or fulltime job (see Table A.6 of the Appendix A).¹⁷ This last number is close to the 4 percentage

 $^{^{15}}$ According to Hughes et al [2011], a "growing literature suggests that for many people, the net health effects of grandchild care are negative" (p.111).

¹⁶See Table A.5 of the Appendix A for the regressions with interaction terms. Estimation results are consistent to the inclusion of a polynomial for age instead of age group dummies as well as to including years of education instead of educational categories. Given that marital status and spouse's income might be correlated with living close to parents or in-laws we also check consistency of results, excluding both variables.

¹⁷For our estimations regarding labor force participation we only consider a woman in the labor force if she works a regular part-or fulltime job. All estimation results are consistent to the inclusion of a polynomial for age instead of age group dummies as well as to including years of education instead of educational categories. Results are robust to the exclusion of the variables marital status and spouse's income. Including interaction terms of living close to parents or in-laws and educational attainment seems to indicate that the positive effect of living close to parents or in-laws on mothers' labor force participation is driven by individuals with tertiary education. However, coefficients are only (negatively)

points lower bound of estimates for the US in Compton and Pollak [2011]. The latter argue that geographical proximity is a good instrument for child care arrangements because its positive effect on labor force participation does not extend to groups for which grandparent-provided child care is not a determinant for labor supply. Similarly, in our data the effect does not extend neither to men nor to single childless women (see Table A.7 of the Appendix A).

2.3 Costs of Grandparent-Provided Child Care

Lower wages Controlling for selection effects, we find that mothers living close to their parents or in-laws earn lower hourly wages.¹⁸ The first column of Table 2.1 displays the coefficients for the Heckman selection model for log hourly wages for mothers in Germany age 25 to 50. Living close to parents or in-laws or in the same house increases the probability of holding a regular part-or fulltime job, but it reduces hourly wages by almost 5%.¹⁹ Other control variables show the expected signs. Wages are higher for those living in larger communities, they increase with firm tenure, and tertiary education. On the other hand, having more children, living in small communities, not being German, and living in East Germany are all aspects that affect hourly wages negatively.²⁰ We also check if our results are driven by low-educated individuals. Running the regression separately for individuals with and those without tertiary education shows that this is not the case. On the contrary, the penalty in hourly wages for staying close to parents or in-laws turns out to be higher – more than 6% – for the group of highly educated individuals (see Tables A.11 and A.12 of the Appendix A).

The two exclusion restrictions that we use are: (i) having a child younger than age three and (ii) spouse's income. Both variables affect mothers' labor force participation (see Table A.6 of the Appendix A). However, none is directly related to mothers' hourly wage rates. Given that we only consider mothers, differences in a child's age are mostly ex-

significant for individuals with primary or secondary education.

¹⁸For this analysis, we only consider wage incomes of dependent workers of regular full-or part time jobs.

¹⁹Using log monthly wages, controlled for by hours worked, leads to slightly more negative coefficients for living close (see Table A.8 of the Appendix A), as does not controlling for selection effects (see Table A.9 of the Appendix A for an OLS regression of log hourly wages).

 $^{^{20}}$ Estimation results are consistent to the inclusion of a polynomial for age instead of age group dummies as well as to including years of education instead of educational categories. We also check the robustness of our results to the exclusion of the variables marital status and spouse's income (see Table A.10 of the Appendix A).

	Log hourly		Selection	
	wage		Equation	
	(1)		(2)	
Married, living together	-0.009	(0.016)	-0.324***	(0.044)
Number of children	-0.037***	(0.010)	-0.286***	(0.017)
Other than German nationality	-0.134*	(0.077)	-0.312**	(0.140)
Tertiary education (ISCED: 5,6)	0.293^{***}	(0.015)	0.441^{***}	(0.036)
Parents or in-laws close	-0.049***	(0.014)	0.073^{**}	(0.033)
Parents or in-laws in same house	-0.050**	(0.020)	0.056	(0.048)
Small community	-0.040**	(0.017)	-0.006	(0.039)
Large community	0.046^{**}	(0.019)	0.045	(0.044)
in East Germany	-0.069	(0.044)	0.294^{**}	(0.119)
Log (Spouse's income) [†]			0.015^{***}	(0.005)
Children 0-2			-1.114***	(0.056)
Tenure in firm	0.016^{***}	(0.001)		
Constant	1.905^{***}	(0.046)	0.188**	(0.087)
Observations	8,129		8,129	

Table 2.1 :	Proximity	to grandpa	arents and	hourly w	ages	
Coefficients	of Heckman	Selection	Model for	mothers'	log hourly	wages

 \dagger Missing values and values < 1 are set to 0. Standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1. Data: SOEP unbalanced panel 91,96, 01,06; Mothers 25-50. Reference group: unmarried mothers of age 25-29 of children age 3 or older living in West Germany, with education level 1 or 2 (ISCED:0-4) in 1991, in a medium-sized West German town in North Rhine-Westphalia, far from parents or in-laws. All regressions include year dummies, age group dummies and state dummies.

plained for by the mother's own age and her years of education. We control for both variables in the wage estimation. The second exclusion restriction might be invalid if assortative matching leads to similar incomes of wife and spouse. As long as these similarities are mostly explained for by a common level of education or the size of the community they live in – both variables that we include as controls – the use of this exclusion restriction is justified.

More Commuting Costs of spatial restrictions may also arise from commutes. Rupert et al [2009], looking at French data find that mothers with small children, who have a lower bargaining power as workers incur in longer commutes. We find that for working mothers of children of age six or younger the probability of having to commute is almost 9 percentage points higher if their children are regularly cared for by relatives. Table 2.2 dis-

plays marginal effects from the probit estimation for the probability of having to commute for mothers in Germany age 25 to $50.^{21}$

Married, living together	0.005	(0.055)
Number of children	0.053	(0.033)
Other than German nationality	-0.278**	(0.127)
Children 0-2	-0.026	(0.063)
Log (Spouse's income)†	-0.003	(0.011)
Tertiary education (ISCED: 5,6)	0.127^{**}	(0.051)
Tenure in firm	0.004	(0.004)
Children cared for by relatives	0.087^{*}	(0.048)
Children in non-relative care	-0.010	(0.072)
Small community	0.262^{***}	(0.056)
Large community	-0.213***	(0.065)
in East Germany	-0.227*	(0.136)
-		. ,
Observations	537	

 Table 2.2: Grandparent-provided child care and commutes

 Marginal effects from Probit estimation for mothers

†Missing values and values < 1 are set to 0. Standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1 Data: SOEP unbalanced panel, 01,06; mothers 25-50 of children <= 6 years with regular full or part time job. Reference Group: unmarried mothers of age 25-29 in 2001 with education level 1 or 2 (ISCED: 0-4) in a medium-sized West German town in North Rhine-Westphalia, whose children are not in nursery, nor cared for by relatives. All regressions include dummies for age groups, states, and years.

Having children in a nursery or with a sitter, on the other hand, does not affect the likelihood of being a commuter. Other control variables show the expected sign. The probability of having to commute is lower for those of other nationalities and for individuals living in large communities in East Germany. It is higher for mothers in small communities.²²

 $^{^{21}\}mathrm{We}$ only consider waves 2001 and 2006 that include a question on commutes.

 $^{^{22}}$ Again results are consistent to the way the variables age and education are included. Having a child in a nursery may be correlated with child care by relatives. Hence, we also check the robustness of our results to the exclusion of this variable (see Table A.13 of the Appendix A). When running separate regressions for individuals with and without tertiary education, coefficients loose significance. However, coefficients for relative-provided child care for both groups of individuals with and without tertiary education are similar (see Table A.14 of the Appendix A).

Given strong interdependencies of decisions regarding residence, fertility, child care arrangements, and female labor force participation, our empirical analysis faces problems of endogeneity and reversed causality. By using geographical proximity between adult children and their parents as an indirect measure of grandparent-provided child care, we addressed the interdependency of child care arrangements and female labor force participation. However, certain caveats remain because we cannot dismiss a reversed causality between geographical proximity and labor market outcomes or fertility. Nevertheless, from the observed relationships between labor force participation, fertility, commuting, and wages with grandparent-provided child care a set of interesting questions arise: How valuable is grandparent-provided child care in terms of fertility and employment? How do family policies compare to grandparent-provided child care with respect to effects on aggregate employment, mobility, and fertility? In order to answer these questions and to better disentangle women's decision, we build a model economy that incorporates the spatial restrictions imposed by grandparent-provided child care.

3 The Model

In our model economy there are two regions where individuals can reside, 'Home' denoted by 'H' and 'Far', denoted by 'F'. The only ex-ante difference between the two regions is that grandparent-provided child care is only available in 'H'. The economy is populated by a continuum of married women of mass one.²³ Women in our economy live for two periods, each of three years. Essentially we want to capture mothers' decisions during the time when child care is most important, i.e. during early childhood.

At the beginning of the first period, women receive two 'life-course offers', one associated with living in 'H' and the other associated with living in 'F'. Each offer consists of: i) a realization of her labor productivity x and ii) an exogenous source of income, z representing a spouse's income, where $x \in x_1, x_2, ..., x_N$ and $z \in z_1, z_2, ..., z_N$, with N denoting the number of different labor productivities and spouses' incomes in the economy. Offers are independent of the region and independent of each other. Upon observing these two

 $^{^{23}}$ In our SOEP sample only 2.4% of single women (heads of household, not living with another adult) are mothers, while 86% of married women who live together with their spouses are mothers. Hence, we only model married women's decisions. Even though marriage and residence decisions might be related, in order to keep the analysis tractable we abstain from modeling a marriage market and we simply assign an exogenous income to each woman to represent her husband's income.

offers, women have to decide where to reside. We assume that residence choices are only taken once during a woman's life time and cannot be reconsidered. Let D denote the residence choice that takes on value 1 if a woman decides to reside in 'F.' If she decides to reside in 'H', D is equal to 0. Women also have to decide whether to have children and how much to work. In the first period, women decide whether or not to have children, k = 1 or k = 0. Children remain with their mother during the two periods. Each period a woman is endowed with one unit of productive time and she has to decide how much to work.

Working mothers need child care. The price of child care per unit of time, $p^i(D)$ depends on the woman's residence choice and on the age of her child, with $i \in 1, 2$. Mothers with small children – from age 0 to 3 – who have no access to free child care pay $p^1(1)$. Mothers with older children – ages 3 to 6 – pay $p^2(1)$. Living in 'H' potentially provides access to free child care by grandparents, $p^i(0) = 0$. However, with a certain probability grandparents fall sick, are not alive, still work, or are otherwise unable to take care of their grandchildren. In the first period, only a share p(g) of women who live in 'H' obtain free child care. The remaining (1 - p(g)) have to purchase child care at price $p^1(1)$. In the second period, only a share p(g) of those who had access to free child care continue to do so. The remaining (1 - p(g)p(g)) face child care costs equal to $p^2(1)$. Women might receive a subsidy for child care ω from the government. In this case they pay $(1 - \omega)p^i(1)$ for each unit of time their children spend in child care.²⁴ Women might also receive family benefits T conditional on having children.

Women with children (k = 1) care about the quality of their children, e and about consumption, c. Hence, they enjoy the following utility

$$U(c, e, k) = \frac{c^{1-\sigma}}{1-\sigma} + \left(\sigma^e e - \sigma^k\right)k,$$

where σ^k are fixed costs per child. Childless women (k = 0) only care about consumption. The quality of children, e is a function of the time a mother spends with her children, t_m and the time her children spend in child care t_c ,

$$e = \phi_m t_m + \phi_c t_c.$$

Time spent in child care is assumed to be equal to the time the mother is at work. A mother spends her remaining time taking care of her children,

$$t_c + t_m = 1.$$

²⁴We assume that this subsidy is only paid to women who purchase child care at price $p^i(1)$, i.e. to those living in 'F' and those living in 'H' who have no access to free grandparent-provided child care.

How a mother divides her time between working and taking care of her children crucially depends on how decisive her time is for her children's quality. This is captured by the two weights, ϕ_m for time spent with the mother and ϕ_c for time spent in child care or with grandparents. We assume time in child care and time spent with grandparents to be of equal importance for children's quality.²⁵

3.1 Value functions

We solve our model backwards by first presenting the value functions for women in the second period.

Value functions in the second period For childless women with labor productivity x and spouse's income z, the value of residing in 'H' is

$$H^2(x,z) = \frac{c^{1-\sigma}}{1-\sigma},$$

subject to the budget constraint

$$c = (1 - \tau)(x + z),$$

where τ denotes an income tax rate. Childless women only care about consumption, they work all their disposable time, l = 1 and they consume all disposable income. The value of residing in 'F' for a childless woman of type x and spouse's income z is equal to the value for living in 'H', $F^2(x, z) = H^2(x, z)$.

When deciding how much to work, mothers, on the other hand take into account that their children's quality depends on how much time they spend taking care of them. Thus

²⁵Formal child care and grandparent-provided child might be very different. For the first nine months, Hansen and Hawkes [2009] find that formal child care is associated with higher school readiness scores, while grandparent-provided child care is associated with a higher vocabulary test score; both measured at the age of three. Hence, it is not clear whether one of them is of higher quality than the other. Bernal and Keane [2011] find that informal child care has negative effects on children's test scores while centerbased care does not. However, the authors study single mothers only. This clearly introduces a bias towards more disadvantaged backgrounds of mothers and as well as towards informal care providers. In another study, Bernal [2008] finds that non-maternal child care is detrimental for children's scores. Hence if grandparent-provided child care is similar to the type of care a mother provides, then grandparents might be the second best option. Moreover, children in informal care might receive more individual attention, see Clarke-Stewart et al. [1994] and grandparents tend to guarantee a stable provider-child relationship, something found to be determinant for the quality of child care (see Walker [1991]).

the value function for a mother living in 'H' who has access to free child care during the second period is given by

$$H^{2}(x, z, k) = \max_{l} \left(\frac{c^{1-\sigma}}{1-\sigma} + (\sigma^{e}e - \sigma^{k})k \right),$$

subject to the budget constraint

$$c = (1 - \tau)(xl + z) + T,$$

and given the children's quality production function

$$e = \phi_m t_m + \phi_c t_c.$$

Women only decide to have children during the first period. At the beginning of the second period all children are three years old. Mothers who work and who have no access to free child care –those living in 'F' and those living in 'H' but without access to grandparent-provided child care – will have to purchase child care at price $p^2(1)$ per unit of time worked. The value function for a mother without access to free child care in the second period is given by

$$F^{2}(x, z, k) = \max_{l} \left(\frac{c^{1-\sigma}}{1-\sigma} + (\sigma^{e}e - \sigma^{k})k \right),$$

subject to the budget constraint

$$c = (1 - \tau)(xl + z) + T - (1 - \omega)p^2(1)l,$$

and given the children's quality production function

$$e = \phi_m t_m + \phi_c t_c.$$

Value functions in the first period A woman who resides in 'H' has to decide whether to have children and how much to work. Only a share of working women p(g) has access to free child care provided by grandparents. If a woman has access to free child care, her value function is given by

$$H^{1}(x,z) = \max_{l,k} \left(\frac{c^{1-\sigma}}{1-\sigma} + (\sigma^{e}e - \sigma^{k})k + \beta(p(g)H^{2}(x,z,k) + (1-p(g))F^{2}(x,z,k) \right),$$

subject to the budget constraint

$$c = (1 - \tau)(xl + z) + T,$$

and given the children's quality production function

$$e = \phi_m t_m + \phi_c t_c.$$

Given that residence choices are only made at the beginning of the first period, the continuation value for a woman living in 'H' is equal to the discounted value of living in 'H' in the second period. In the second period, with probability p(g), she continues to have access to grandparent-provided child care. On the other hand, a share (1 - p(g)) of mothers who live in 'H' and who work, have to pay for child care. Their problem is the same as that of working mothers living in 'F'.

If a woman resides in 'F', she has to decide whether to have children and how much to work, taking into account that if she has children and works, she has to purchase child care at price $p^1(1)$. Child care services might be subsidized at a rate ω . The value function in the first period for a woman living in 'F' or living in 'H' but without access to free child care is given by

$$F^{1}(x,z) = \max_{l,k} \left(\frac{c^{1-\sigma}}{1-\sigma} + (\sigma^{e}e - \sigma^{k})k + \beta F^{2}(x,z,k) \right)$$

subject to the budget constraint

 $c = (1 - \tau)(xl + z) + T - (1 - \omega)p^{1}(1)l$

and given the children's quality production function

$$e = \phi_m t_m + \phi_c t_c.$$

Residence Decisions When deciding where to reside, women do not know for sure if they will have access to grandparent-provided child care in 'H'. They thus consider the expected value of living in 'H' which is given by

$$EH^{1*} = p(g)H^{1*}(x,z) + (1-p(g))F^{1*}(x,z)),$$

where $F^{1*}(x, z)$ and $H^{1*}(x, z)$ denote the value functions evaluated at the optimal decisions of labor supply and number of children. Women decide where to reside by comparing the expected value of living in 'H' to the value of living in 'F'. They will decide to live in 'F' if the expected value of living in 'H' is strictly higher than the value of living in 'F'.²⁶

²⁶Individuals who are indifferent are distributed equally across the two regions.

If child care were free in both regions there would only be two reasons why women would decide to reside in 'F': (i) a higher labor productivity and/or (ii) a higher spouse's income. However, once child care is costly women's residence choices depend on the life-course offers for each of the two regions, the price of child care in 'F' and the availability of free child care in 'H'. A woman who receives the same life-course offer in both regions- the same productivity x and the same spouse's income z – will always decide to live in 'H' because this is where she may obtain free child care. However, life-course offers can differ. Assume that a woman receives an offer $(x, z)^H$ associated with living in 'H' and an offer $(x,z)^F$ associated with living in 'F'. Her offer could include a higher labor productivity in 'F', $(x^F > x^H)$, a lower labor productivity in 'F', $(x^F < x^H)$ or labor productivities could be the same, $(x^F = x^H)$. Moreover, spouses' incomes might also be different. In this case, residence choices become non-trivial. A higher offer might not be enough to compensate for the cost of child care and a woman might decide to reside in 'H' even if the life-course offer in 'F' is higher. In the following subsection we analyze the closed form solution for labor supply in our model and we discuss how changes in the price of child care and life-course offers affect women's labor supply decision.

3.2 Labor Supply

For our relatively simple model of grandparent-provided child care and women's labor market outcomes, we are able to obtain a closed-form solution for the optimal labor supply. Childless women work all their disposable time, $l^* = 1$. For mothers on the other hand, the optimal labor supply is given by

$$l^* = \left(\frac{1}{\sigma^e(\phi_m - \phi_c)}\right)^{\frac{1}{\sigma}} \left(x(1-\tau) - (1-\omega)p^i(1)\right)^{\frac{1-\sigma}{\sigma}} - \frac{(1-\tau)z + T}{(1-\tau)x - (1-\omega)p^i(1)}$$

subject to

$$0 \le l^* \le 1$$
, for $i = 1, 2$.

Mothers' optimal labor supply depends crucially on how important time spent with their mother is for children's quality, compared to the time spent in child care or with grandparents. We assume that time with mothers is at least as important as time in child care $(\phi_m \ge \phi_c)$. Mothers' labor supply also depends on the relationship between mothers' labor productivity and the cost of child care, as well as on the relative value of the spouse's income compared to mothers' labor productivity. Women whose marginal benefit from working – their labor productivity x – is lower than the marginal cost of working – the cost of child care, $p^i(1)$ for $i \in 1, 2$ – will decide to stay home. On the other hand, if the mothers' time for children's quality were of equal importance as time spent in child care, $\phi_m = \phi_c$, women whose labor productivity is high enough would work all their disposable time

$$l^* = \begin{cases} 0 & \text{for } (1-\tau)x < (1-\omega)p^i(1) & \text{and } \phi_m \ge \phi_c \\ 1 & \text{for } (1-\tau)x > (1-\omega)p^i(1) & \text{and } \phi_m = \phi_c. \end{cases}$$

In case $\phi_m > \phi_c$, women whose labor productivity suffices to pay for child care will work, $0 \leq l^* \leq 1$. Even though, a mother's time has a higher weight for children's quality, working increases disposable income and consequently consumption. The effect of other parameters on the optimal labor supply is as expected. A higher weight of consumption for utility (σ), increases labor supply. An increase in the importance of time spent in child care for children's quality and an increase in labor productivity have similar effects. On the other hand, an increase in the weight of a mother's time for children's quality reduces labor supply. Similarly an increase in the cost of child care and an increase in the spouse's income decrease labor supply. In particular mothers supply no labor if their spouse's income is large enough,

$$l^* = \{ 0 \text{ if } z > \frac{1}{1-\tau} \left[\left(\frac{x(1-\tau) - (1-\omega)p^j(1)}{\sigma^e(\phi_m - \phi_c)} \right)^{1/\sigma} - T \right].$$

4 Calibration Strategy

In order to be able to quantify the importance of grandparent-provided child care and the effect of family policies for women's decisions on residence, labor supply, and fertility, we calibrate our model. Some parameters of the model are fixed based on available evidence. The remaining parameters are calibrated to match the model to several labor market statistics and fertility data from Germany. Most statistics used for calibration come from pooled waves 1991, 1996, 2001, and 2006 of the German Socio-Economic Panel (SOEP). We consider weighted statistics for married women in West Germany ages 20 to 48 for whom information on parents' residence is available.²⁷ Finally, we set policy parameters to represent German family policies.

In our model economy, individuals are distributed over different life-course offers. This distribution is denoted by $\Pi(x^h, z^h, x^f, z^f)$, chosen such as to be consistent with the existence of an initial distribution of women over labor productivities, $\Omega(x)$ and an initial distribution of men over labor productivities, $\Theta(z)$. Hence $\Omega(x_i)$ denotes the mass of women who are of productivity type x_i , for i = 1...N. Labor productivities x and z follow a log normal distribution. We discretize the distribution of men's observed wages to

 $^{^{27}}$ We chose ages 20 to 48 because this is the age range of women with children 0-6. We only consider individuals who report if they have children, and if and how they participate in the labor market. For men, we use wage data for married working individuals older than 20.

obtain twenty different types of spouses' incomes. Mean and standard deviation of men's productivity distribution are denoted by μ_z and σ_z . The distribution of women's observed wages, on the other hand, is much more likely to be effected by selection into employment. We assume the standard deviation of men and women's wages to be the same and we calibrate the mean of women's underlying productivity distribution to match the observed mean wage for women. Estimates for these parameters $(\mu_z, \sigma_z, \sigma_x)$ and calibration targets (μ_x) are taken from the distribution of hourly wage rates observed in our SOEP sample. They are $\mu_z = 2.74$ and $\sigma_z = 0.41$ for men and $\mu_x = 2.46$ and $\sigma_x = \sigma_z = 0.41$ for women.

A life-course offer is a labor productivity and a spouse's income. It can be interpreted as a couple formed by a woman of productivity, x and a man of income z. Thus, the probability of an offer is equivalent to a matching probability between a woman and a man. The probability that an offer $(x_i, z_j)^d$ for d = H, F is realized, or that a woman of type x_i matches with a man of type z_j is denoted by $\Phi(x, z)$. If a man and a woman are of the same type, i = j, this probability is equal to ψ . Otherwise the probability is given by $\frac{1-\psi}{N-1}$ for $i \in 1, ..., N$ and $j \in 1, ..., N$. To assign a value to ψ , we use the degree of assortative matching indicating how likely it is to meet your own productivity type. Fernández et al. [2005] estimate this value to be 0.7 for Germany, i.e. 70% of women in Germany match with men of the same type. The remaining 30% are equally likely to match with men of types different from their own.

In order to determine the distribution of life-course offers, $\Pi[(x, z)^H, (x, z)^F]$, we need to know how many individuals receive an offer $(x_i, z_j)^H$ in 'H' and an offer $(x_r, z_p)^F$ in 'F', for $i, j, r, p \in 1, ..., N$. For instance, how many women receive a particular offer $(x_1, z_3)^H$ in 'H' and an offer $(x_3, z_5)^F$ in 'F'? The number of individuals receiving the offer $(x_1, z_3)^H$ depends on the number of individuals of type x_1 – given by $\Omega(x_1)$ – and on the number of individuals of type z_3 , – given by $\Theta(z_3)$ – and on the probability of this match happening, $\Phi(x_1, z_3)$. Thus the number of women receiving an offer $(x_1, z_3)^H$ is equal to the product of the three elements, $\Omega(x_1)\Theta(z_3)\Phi(x_1,z_3)$. It is thus equal to the probability of a woman being of type x_1 and the probability of a man having income z_3 and the probability for this match to happen. Likewise, the number of individuals receiving an offer $(x_3, z_5)^F$ is given by $\Omega(x_3)\Theta(z_5)\Phi(x_3,z_5)$. Hence the number of individuals receiving the two offers detailed above is equal to the probability of individuals receiving an offer $(x_1, z_3)^H$ times the probability of individuals receiving an offer $(x_3, z_5)^F$, normalized by the probability of receiving any offer. Given that there is a mass one of individuals, the number of individuals of a certain type is given by the probability of being an individual of that type. Each element of $\Pi[(x,z)^h, (x,z)^f]$ is calculated in the same fashion. The distribution of life-course offers is key to our analysis, as it determines how individuals are distributed across different offers, which will have implications for the model's aggregate statistics.

According to SHARE around 16.8% of German grandparents who live close to their small grandchildren [0-3), take care of them on a daily basis. Mothers of around 89% of women in our SOEP sample are still alive. Hence we assign value 0.14 to the parameter p(g) that denotes the share of women who live in 'H' and who have access to grandparent-provided child care. Given that one model period is equivalent to 3 years, the discount factor, β is set to a value of 0.889 in order to match a yearly interest rate of 4%. Table 4.3 displays all parameters set a priori.

Table 4.3: Parameters based on a priori information

Parameter	Explanation	Value
σ_x	standard deviation of women's log productivity	0.41
μ_z	mean log productivity of spouses	2.74
σ_z	standard deviation of spouses' log productivity	0.41
p(g)	% of women with access to free care in 'H'	0.14
ψ	assortative matching parameter	0.7
β	discount factor	0.889

We calibrate the parameters of the utility function, $(\sigma, \sigma_e, \sigma_k)$, the cost of child care in both periods, $p^1(1)$ and $p^2(1)$, women's mean log hourly wage rate μ_x , as well as the parameters of the children's quality function, (ϕ_m, ϕ_c) . Note that we impose $\phi_m + \phi_c = 1$, and hence we only need to calibrate one of the two parameters. We now relate the calibrated parameters to the data moments that they are most likely to affect.

We set the mean of the distribution of women's underlying productivities \bar{x} to 2.31 in order to match working women's observed mean log hourly wage rate, μ_x of 2.46. The weight of consumption in the utility function, σ is set to 0.976, to match the percentage of working women in 'H'.²⁸ Given that on average these women face lower child care costs, their participation decision is very much determined by the value of consumption. We use two moments related to fertility and participation to match the fixed utility cost of children, σ_k and the weight of children's quality, σ_e in the utility function. The weight of children's quality affects the decision of whether to participate in the labor market or not. Thus, to match this parameters we use the participation rate of mothers with older

²⁸We join variables "parents or in-laws in same house" and "parents or in-laws close." Hence those in 'H' are individuals who live in the same house, neighborhood, or town as their parents or in-laws.

children [3-6) in 'H' who face lower child care costs. Whether a woman wants to become a mother or not is related to the fixed cost of children. Hence we use the percentage of women who are mothers in 'H' to match the fixed cost σ_k . We set σ_k and σ_e to 0.35 and 1.2 respectively.

The weight of mothers' time for children's quality, ϕ_m determines how much a mother works depending on her child's age. We calibrate this parameter to match the percentage of working mothers with small children [0-3) in 'H'. This parameter takes on value 0.672. The OECD [2008] estimates that child care costs for a two-year-old in Germany amount to 9.1% of average income. We set the first period child care cost for small children [0,3) $p^1(1)$ to 1 to match this number.²⁹ The price of child care in the second period is calibrated to match the participation rate of mothers with older children [3-6) in 'F'. This parameter takes on value 0.75. Table 4.4 displays the calibrated parameters of the model.

Table 4.4:	Calibrated	parameters
------------	------------	------------

Parameter	Explanation	Value
\overline{x}	mean underlying productivity of women	2.31
σ	weight of consumption	0.976
σ_e	weight of children quality	1.2
σ_k	fixed cost of children	0.35
ϕ_m	weight of mother's time	0.672
$p^{1}(1)$	cost of child care, period 1	1
$p^{2}(1)$	cost of child care, period 2	0.75

Finally, the model's policy parameters are the income tax rate, τ , child care subsidies, ω and family benefits, T. We set τ to 37% which is equivalent to the income tax revenue collected by the German government as a fraction of GDP (OECD [2010]). According to the OECD [2009], all German families receive some family benefits for each child up to the age of eighteen (*Kindergeld*). In particular, they receive 184 Euros per month for the first child, 190 for the second and 205 for the third, fourth, fifth child etc. We set the amount of family benefits in our economy, T such as to match that for the average German family (in terms of income and number of children), *Kindergeld* is equivalent to

²⁹Note that in our model men and women who are not taking care of their children work all their disposable time. Given that we targeted hourly wage rates, the resulting average income in the model is too high compared to the data. In order to calculate child care costs as a fraction of average income we impose a maximum work day equal to 40% of disposable time, i.e. 40 hours per week of 98 hours disposable time after subtracting 10 hours per day for sleeping, eating, personal care.

around 5% of family income. Hence, T is set to 0.522. According to the same source, child care subsidies for small children [0-3) are negligible in Germany. Therefore, we set child care subsidies (ω) to zero. All policy parameters are displayed in Table 4.5.

Parameter	Explanation	Value
${ au \over T} \omega$	income tax Family Benefits child care subsidy	$0.37 \\ 0.522 \\ 0$

Table 4.5: Policy parameters

5 Results-Benchmark Economy

Table 5.6 presents model moments from our benchmark economy together with the corresponding data moments from our SOEP sample.

Table 5.6:	Data	and	model	moments

	Data	Model
LFP of mothers, children [0-3) in 'H'	26.75	28.68
LFP of mothers, children $[3-6)$ in 'H'	46.22	51.09
LFP of mothers, children $[3-6)$ in 'F'	44.10	49.17
LFP of women in 'H'	50.36	57.86
Mean log hourly wage rate of working women	2.46	2.46
% of women being mothers in 'H'	72.34	70.09
Child care costs as $\%$ of average income	9.10	9.88

Our model matches labor force participation of mothers with small children [0-3) who live close to parents or in-laws, particularly well. This rate is equal to 26.75% in the data while the model estimates it to be 28.68%. On the other hand, the model slightly overestimates labor force participation of mothers with children between the ages of 3 and 6. While in the data this rate is 46.22% in 'H', the model estimates a rate of 51.09%.

In 'F', this rate is 44.10% in the data while the model predicts that 49.17% of mothers with children between 3 and 6 who live far from parents or in-laws work. We slightly overestimate the labor force participation of all women who live close to parents or in-laws. This rate is 50.36 in the data, while the model estimates a rate of 57.74. This is partly due to overestimating labor force participation of mothers with older children in 'H'. In addition, all childless women work all their disposable time because leisure has no value for utility in our model set-up. The model predicts an average log hourly wage rate for working women of 2.46, matching perfectly the wage rate observed in the data. The model also performs fairly well in terms of matching data on fertility. In our model, women can only decide whether to become mothers or not. Thus we cannot say anything about the average number of children per woman. However, the model matches closely the percentage of married women who are mothers, 70.09% in the model compared to 72.34% in the data. In Germany, the cost of child care for small children [0-3), is equal to 9.10% of average income while the model predicts this cost to be 9.88%. Thus the model also matches child care costs well.

Model moments in Table 5.6 were targeted explicitly to calibrate certain parameters. In order to assess the model's validity for carrying out policy analysis, we need to consider the model's performance in matching moments that have not been used for calibration. Table 5.7 shows these un-targeted moments of the model and the corresponding data moments.

Table 5.7	: Data	and	model	moments:	not	used	for	calibration
-----------	--------	-----	-------	----------	----------------------	------	-----	-------------

	Data	Model
% of women being mothers, 'F'	66.75	64.28
Aggregate LFP rate of women	48.90	57.74
LFP rate of women, 'F'	47.62	57.61
LFP rate of mothers children $[0-3)$, 'F'	20.55	19.00
Share of female married population living in 'H'	43.43	52.62
Average time spent with child, all mothers, as $\%$ of total time	64.37	57.14
Average time spent with child, working mothers, as $\%$ of total time	(35.33-57.24)	37.64

The model produces particularly good estimates of the percentage of women who are mothers in 'F' as well as of the labor force participation of mothers with small children in 'F'. These statistics are key to our analysis of the relationship between mothers' labor force participation and the availability of child care. On the other hand, as expected our model without leisure in the utility function overestimates aggregate female labor force participation. Moreover, the model predicts that 52.62 per cent of women live in 'H', while in the data, 43 per cent of women live close to parents or in-laws. The model somewhat overestimates this statistic. However, taking into account that we did not target this number in the calibration the model does fairly well.

We also report the percentage of time mothers devote to taking care of their children. In our model, individuals do not value leisure and hence when a mother is not working, she is taking care of her children. Hence, time spent taking care of children is a residual. However, in the data we observe women spending their non work-related time on other activities besides child care. Hence, we cannot reproduce actual time spent on child care, but we want to show how our model performs in terms of mother's time spent on child care relative to work-related activities. In order to do so, we make the data comparable to our model by assuming that the sum of time women spend on child care and time for work-related activities is equal to their total disposable time. Our data comes from an OECD report on time use, OECD [2011]. It contains the time women in a couple with at least one child of preschool age devote to various activities. We are interested in how much time women spend on child care—both as primary or secondary activity- as a percentage of total time- child care plus work-related time. For Germany, all mothers spend 63.37 per cent of their time on child care, while our model predicts that all mothers spend 57.17 per cent of their time on child care. For working women, the data differentiates between those working part time and those who work full time. We report both values and we take them to be lower and upper bounds for time spent taking care of children by working mothers in our model. In the data, mothers who work full time spend 35.33 per cent of their time caring for their children. Meanwhile those who work part time spend a larger share of their time taking care of their children, 57.24%. In our model, working mothers spend 37.64% of their time on child care. This percentage of time spent on child care is relatively close to that of full time working mothers in the data. Despite the lack of leisure in the model and if it were to be the case that women spend time only at work or taking care of children, the model produces sensible moments on mothers' time taking care of children.

6 Discussion

6.1 Cost in terms of wages

In the empirical analysis of the paper we showed that there exists a negative relationship for mothers between living close to parents or in-laws and hourly wages. In our model, where residence is a choice and depends on life-course offers received and on child care costs, the same relationship arises. Figure 6.2 shows the wage distributions of working mothers in both regions. We obverse that the mean of the wage distribution in 'F' (15.5) is higher than in 'H' (13.5). The model thus generates endogenous differences in wage rates of working mothers. In our model differences in child care costs generate these differences in wage rates. In 'F', there is no access to grandparent-provided child care, and women who work have to pay for child care. This leads to a selection of high productivity type working women into residing in 'F'.





In our empirical analysis, we used a Heckman selection model to control for the fact that observed wages might be biased upward by a selection of women into employment. However, we were not able to control for a selection into residing close or far from parents or in-laws. In the model on the other hand, we are able to also control for this additional selection effect. Figure 6.3 displays the underlying or potential wage distributions for

women in both regions, before employment decisions. Women reside in 'F' to take advantage of higher wages. Thus we observe that the potential wage distribution of women in 'F' lies to the right of the one in 'H'. To analyze whether those who reside in 'H' despite lower average wages do so because of grandparent-provided child care, we consider the underlying wage distributions in both regions when there are no grandparents. Figure 6.4 shows that in this case, wage distributions are the same in both regions. In addition to the effect of selection into employment, residence choices also contribute to the differences in the wage distributions of women in 'H' and in 'F'. While these differences are small they clearly depend on how wide-spread grandparent-provided child care is.



Figure 6.3: Women's potential wage distribution

In our model, women decide where to reside, based on the life-course offers they receive and taking into account the higher costs of child care in 'F'. Life-course offers consist of different wages and different spouses' incomes in both regions. In order to better analyze the role of life-course offers for women's decisions, we consider life-course offers that only differ in one of the two dimensions. First we analyze the role of women's wages by only allowing offers to differ across women's wages in each region. In this case, a life-course offer consists of a wage and spouse's income in 'H' and a different wage and the same spouse's income in 'F'. In a second step we maintain wage offers fixed and only vary spouses' incomes across regions.



Figure 6.4: Women's potential wage distribution, no grandparents

6.2 Effect of women's wages on decisions

To study the importance of women's wages for residence choices, we shut down the spouse's income channel. In this case, the only reason to move to 'F' is a higher wage rate. The wage distribution of women in 'F' is thus skewed to the right (see Figure 6.5). The distribution of spouses' incomes in 'F', on the other hand is fairly symmetric. Only 22% of women live in 'F' compared to 48% in the benchmark economy. Without any variation in spouse's income 58% fewer women reside in 'F'. In our benchmark economy with a certain probability – as specified in the calibration section – a woman could receive life-course offers $(x_1, z_3)^H$ and $(x_1, z_5)^F$. Given certain parameters she would be likely to choose the second option. In the observed distribution of spouse's incomes only z_5 would be registered. When shutting down the spouse's income channel the aforementioned life-course offer will not be available. Instead a women might receive offers $(x_1, z_3)^H$ and $(x_1, z_3)^F$ or $(x_1, z_5)^H$ and $(x_1, z_5)^F$. In one case z_3 and in the other z_5 will become observations in the spouse's income distribution. This is why on the aggregate compared to our benchmark economy women receive lower spouses' incomes and they thus participate more. In particular, in 'F' 70% of mothers with small children participate. These women can afford to pay for child care but would decide not to work if their spouse had a higher income.



Figure 6.5: Women's potential wage distribution and spouses' income distribution

6.3 Effect of spouse's income on decisions

If on the other hand we shut down the wage channel, women receive the same labor productivity offer in both regions, but different offers for spouses' income. In this case, the main reason for moving to 'F' lies in a higher spouse's income and women with relatively higher spouse's income reside in 'F'. The distribution of productivity types in 'F' displays a larger variance than the distribution in 'H' and the distribution of spouses' incomes in 'F' is skewed to the right (see Figure 6.6). Again, fewer women reside in 'F' compared to the benchmark economy, 21% versus 48%. For the same reason that women receive lower spouses' incomes when we shut down the spouse's income channel, they receive lower productivity offers when we shut down the wage channel. We thus also observe a decrease in female labor force participation as women face lower wages on average.

Both channels - women's wages and their spouses' income - are important for understanding women's residence choices as well as their decisions regarding labor force participation. If we shut down both channels and women received the same life-course offers in 'F' and 'H' all women would reside in 'H'. Given a positive probability for grandparent-provided



Figure 6.6: Women's potential wage distribution and spouses' income distribution

child care, the value function for living in 'H' would always be larger than the one for living in 'F'.

7 Counterfactual Experiments

In our first counterfactual experiment we analyze a situation in which grandparents are not available to take care of their grandchildren and in which everyone has to pay for child care. Our second counterfactual experiment considers subsidies for paid child care. The purpose of our first experiment is to quantify the importance of grandparent-provided child care. There are several reasons why the provision of child care by grandparents might be reduced in the future. Women's age at first child birth has increased over the last decades. In 2009, German mothers were on average 30 years old when giving birth to their first child, while in 1970 average age at first birth was 24 (OECD [2008]). As successive generations of women delay birth, grandparents may be to be too old or too sick to take of their grandchildren. On the other hand, women's labor force participation and individuals' retirement age has increased in many countries. Situations where both generations of women – grandmothers and adult daughters – are of working age when grandchildren come along are going to be even more likely in the future.³⁰ Regarding our second experiment, we consider a subsidy for child care. Given the positive relationship found in the literature between availability and low cost of child care and female labor force participation, we would expect such a policy to lead to an increase in mothers' labor force participation. We set the subsidy to 53 per cent of child care costs. This is the amount of subsidy needed to reduce the cost of child care in our benchmark model such as to be equal to the cost that Swedish families face. In Sweden, child care costs are among the lowest compared to other OECD countries, and they amount to 4.6 per cent of average income (OECD [2008]).

No grandparent-provided child care Table 7.8 provides moments from our first counterfactual experiment together with the corresponding moments from our benchmark economy.

	No	Benchmark
	grandparents	economy
Aggregate LFP rate of women	56.40	57.74
% of women being mothers, 'H'	65.21	70.09
% of women being mothers, 'F'	65.21	64.28
LFP rate of mothers children [0-3), 'H'	17.85	28.68
LFP rate of mothers children [3-6), 'H'	48.40	51.09
LFP rate of mothers children [0-3), 'F'	17.85	19.00
LFP rate of mothers children [3-6), 'F'	48.40	49.17
share of women living in 'H'	50.00	52.62
-		

Table 7.8: No grandparent-provided child care

In the benchmark economy, women in 'F' and those women in 'H' who have no access to grandparent-provided child care face high child care costs. Without the availability of grandparents, the incentive to remain close disappears. We thus observe that the share of women living in 'F' is 2.56 percentage points higher compared to the benchmark economy.

³⁰Individuals' increasing life expectancy and better health might counteract the problem of late first child birth. However, this increased life expectancy is again likely to lead to successive increases in retirement age.

We observe that half the population chooses to live in 'F' and the other half chooses to live in 'H'.³¹ In both regions 65.21 per cent of women are mothers. Labor force participation of mothers with small children [0-3) decreases in both regions to 17.85 per cent. The fact that grandparent-provided child care is not available in 'H' alters incentives to move to 'F'. This changes the composition of productivity types in 'F' which in turn affects the labor force participation rate of women in 'F'. Given that without grandparents, child care for older children [3-6) is more costly, too we also observe a decrease in labor force participation of mothers with older children.

On the aggregate, labor force participation by women drops 1 percentage point compared to our benchmark economy. Finally, we observe a drop in fertility. In 'H' the number of women who become mothers is reduced by almost 5 percentage points. Women who would decide to reside in 'H' if there was grandparent-provided child care, react by moving more, participating less, and having fewer children. However, given that in our benchmark economy only 14% of grandparents are readily available for child care, effects of a loss in availability of grandparents are small on the aggregate.

Child Care Subsidies In our second counterfactual experiment we consider an increase in child care subsidies. In particular, we consider a policy that subsidizes 53% of child care costs in both periods, i.e. the policy parameter ω is set to 0.53. Table 7.9 displays moments from this counterfactual experiment, next to moments from our benchmark economy, with $\omega = 0$.

Under this policy, every woman can afford to become a mother and around 84 per cent of women work. A child care subsidy not only affects women who live in 'F', but also those who live in 'H' and who do not have access to grandparent-provided child care. It thus increases labor force participation of mothers in 'H' as well as of mothers in 'F'. Some women still have access to free grandparent-provided child care, leading to a slightly higher participation rate of mothers in 'H'. It is for the same reason that we do not observe any change in the share of women living in 'F'. The significant increases in both the aggregate number of mothers and mothers' labor force participation rates, indicate that subsidizing child care costs can lead to higher fertility and higher labor force participation by mothers.

 $^{^{31}}$ The fact that the population splits into half has to do with how the offers are set up and that we equally divide those who are indifferent.

Table 7.9: Child care subsidy, $\omega = 0.53$

		Benchmark
	$\omega=0.53$	economy
Aggregate LFP rate of women	82.91	57.74
% of women being mothers, 'H'	100.00	70.09
% of women being mothers, 'F'	100.00	64.28
LFP rate of mothers children [0-3), 'H'	82.70	28.68
LFP rate of mothers children [3-6), 'H'	84.22	51.09
LFP rate of mothers children [0-3), 'F'	81.51	19.00
LFP rate of mothers children [3-6), 'F'	83.10	49.17
share of women living in 'H'	52.46	52.62
Ŭ		

8 Conclusion

In this paper we document benefits and costs of grandparent-provided child care. Looking at German data we find that women residing close to parents or in-laws are more likely to have children and that as mothers they are more likely to hold a regular full-or part time job. However, we find that their wages are lower and that they are more likely to incur in commutes. We build a model of residence choice, fertility decisions, and female labor force participation to account for this trade-off. We simulate the model to match the German economy in terms of fertility and female labor force participation. We then perform two counterfactual experiments to analyze how women's decisions on residence, fertility, and labor force participation change under distinct scenarios regarding availability of grandparent-provided child care and different family policies. We find that if there was no grandparent-provided child care, fewer women would participate in the labor market and fewer would become mothers. We also observe that mobility increases. Hence, grandparent-provided child care imposes a geographical restriction. One way to remove this restriction would be to subsidize child care. However, we find no effect on mobility when we subsidize 53% of child care costs. Though this subsidy leads to increases in both, aggregate female labor force participation and fertility.

In absence of child care subsidies, grandparent-provided child care plays an important role for mothers' labor force participation. However, women who remain close to their parents or in-laws only access a confined labor market, which might imply worse labor market opportunities and thus lower wages. Hence, when designing policies aimed at increasing labor force participation of mothers, policy makers should take into account the wide-spread presence of grandparent-provided child care as well as the spatial restrictions it implies. For instance, a policy that subsidizes grandparents' time may not only affect mothers' labor force participation but it may also have an impact on their wages and/or commutes.

In this paper we simply assume that being close to one's grandparents' implies that with a certain probability grandparents take care of their grandchildren, and we do not consider grandparents' decisions to provide care. However, this decision might be very much related to individuals' retirement age and especially in the case of grandmothers to previous decisions about labor force participation. In this sense, opposing cohort effects of female labor force participation could arise. On the one hand, having a working mother increases the likelihood for her daughters to also become working mothers.³² On the other hand, a grandmother who is actively participating in the labor market might be less likely to provide child care for her grandchild, something that is found to increase mother's labor force participation rates. We believe that a very interesting road for future research could be an analysis of how late first birth and improved health after retirement interact with these forces.

References

Alesina, Alberto and Paola Giuliano (2010): "The Power of the Family," *Journal of Economic Growth*, 15, pp 93-125.

Attanasio, Orazio; Low, Hamish and Virginia Sanchez-Marcos (2010): "Explaining Changes in Female Labor Supply in a Life-Cycle Model," American Economic Review, 98(4), pp 1517-52.

Arpino, Bruno; Pronzato, Chiara D. and Lara P. Tavares (2010): "All in the family: informal childcare and mothers' labour market participation," Centre for Household Income, Labour and Demographic Economics, Working Paper No. 12/2010.

Bernal Raquel (2008): "The Effect of Maternal Employment and Child Care on Children's Cognitive Development," *International Economic Review*, 49(4), pp 1173-1209.

 $^{^{32}}$ See Fernández et al [2004] and Farre and Vella [2007] on the intergenerational transmission of attitude towards the role of women in the economy and its effect on female labor force participation.

Bernal Raquel and Michael P. Keane (2011): "Child Care Choices and Children's Cognitive Achievement: The Case of Single Mothers," *Journal of Labor Economics*, 29(3), pp 459-512.

Bick, **Alexander** (2010): "The Quantitative Role of Child Care for Female Labor Force Participation and Fertility," mimeo Goethe University Frankfurt.

Blau, David M. and Philip K. Robins (1989): "Fertility, Employment, and Child-Care Costs," *Demography*, 26, (2), pp 287-299.

Bundesministerium für Familie, Senioren, Frauen und Jugend (2005): Gender-Datenreport, 1. Datenreport zur Gleichstellung von Frauen und Männern in der Bundesrepublik Deutschland, München.

Cardia, Emanuela and Serena Ng (2003): "Intergenerational time transfers and childcare," *Review of Economic Dynamics*, 6, pp 431-454.

Clarke-Stewart, K. A.; Gruber, C. P., and L. M. Fitzgerald (1994): "Children at home and in day care," *Hillsdale, NJ: Erlbaum*.

Compton, Janice and Robert A. Pollak (2011): "Family Proximity, Childcare, and Women's Labor Force Attachment," NBER Working Paper 17678.

Dimova, Ralitza and François-Charles Wolff (2011): "Do downward private transfers enhance maternal labor supply? Evidence from around Europe," *Journal of Population Economics*, 24(3), pp 911-933.

DIW (2009): "Real Wages in Germany, Numerous Years of Decline," Weekly Report, Deutsches Institut für Wirtschaftsforschung, Berlin, 28/2009.

del Boca, Daniela (2002): "The effect of child care and part time opportunities on participation and fertility decisions in Italy," *Journal of Population Economics*, 15, pp 549-573.

del Boca, Daniela and Daniela Vuri (2007): "The mismatch between employment and child care in Italy: the impact of rationing," *Journal of Population Economics*, 20, pp 805-832.

Erosa, Andrés; Fuster, Luisa and Diego Restuccia (2010): "A general equilibrium analysis of parental leave policies," *Review of Economic Dynamics*, 13(4), pp 742-758.

European Commission (2009): "The provison of childcare services. A comparative review of 30 European countries," European Commission's Expert Group on Gender and Employment Issues (EGGE).

Farré, L. and F. Vella (2007): "The Intergenerational Transmission of Gender Role Attitudes and Its Implications for Female Labor Force Participation," *IZA discussion paper*, no. 2802.

Fernández, Raquel; Flogi, A. and Claudia Olivetti (2004): "Mothers and Sons: Preference Formation and Female Labor Force Dynamics," *Quarterly Journal* of Economics, 119), pp 1249-99.

Fernández, Raquel; Guner, Nezih and John Knowles (2005): "Love and Money: A Theoretical and Empirical Analysis of Household Sorting and Inequality," *The Quarterly Journal of Economics*, 120(1), pp 273-344.

García-Morán, Eva (2010): "Childcare Costs, Female Employment, and Public Policy," mimeo Universidad Carlos III de Madrid.

Greenwood, Jeremy; Guner, Nezih and John A. Knowles (2000): "Women on Welfare: A Macroeconomic Analysis," *The American Economic Review*, 90(2), Papers and Proceedings of the One Hundred Twelfth Annual Meeting of the American Economic Association, pp 383-388.

Guner, Nezih and John A. Knowles (2009): "Why is the Rate of Single Parenthood Lower in Canada than in the U.S.? A Dynamic Equilibrium Analysis of Welfare Policies," *Canadian Journal of Economics*, 42(1), pp 56-89.

Hansen, Kristine and Denise Hawkes (2009): "Early Childcare and Child Development," *Journal of Social Policy*, 38 (2), pp 211-239.

Holdswoth, Clare and Angela Dale (2009): "Proximity and Coresidence of Adult Children and their Parents: Description and Correlates," WP 2009-215, University of Michigan Retirement Research Center.

Hughes, Mary Elizabeth; Waite Linda J.; LaPierre, Tracey A. and Ye Lou (2007): "All in the Family: The Impact of Caring for Grandchildren on Grandparents' Health," *Journals of Gerontology: Series B*, 62(2), pp 108-119.

Konrad, Kai A.; Kunemund, Harald; Lommerud, Kjell Erik and Julio R. Robledo (2002): "Geography of the Family," *American Economic Review*, 92(4), pp 981-998.

Mendez, Ildefonso (2008): "Intergenerational Time Transfers and Internal Migration: Accounting for Low Spatial Mobility in Southern Europe," MPRA Paper No. 8654.

Mincer, Jacob (1978): "Family Migration Decision" Journal of Political Economy, 86(5), pp 749-773.

Mörck, Eva Anna Sjögren and Helena Svaleryd (2009): "Cheaper Child Care, More Children," IZA Discussion Paper No.3942.

OECD (2001): Employment Outlook 2001, Organisation for Economic Co-Operation and Development, Paris.

OECD (2008): Family Database, Directorate for Employment, Labour and Social Affairs, Organisation for Economic Co-Operation and Development, Paris.

OECD (2009): "Germany Country chapter," Benefits and Wages, OECD series, Organisation for Economic Co-Operation and Development, Paris.

OECD (2010): "Country statistical profile: Germany 2010," OECDiLibrary, Organisation for Economic Co-Operation and Development, Paris.

OECD (2011): "LMF2.5: Time use for work, care and other day-to-day activities,",OECD Family database, Directorate for Employment, Labour and Social Affairs, Organisation for Economic Co-Operation and Development, Paris.

Overturf Johnson, Julia (2005): "Who's Minding the Kids? Child Care Arrangements: Winter 2002," Current Population Reports, U.S Census Bureau, pp 70-101.

Posadas, Josefina and Marian Vidal-Fernández (2012): "Grandparents' Childcare and Female Labor Force Participation," IZA Discussion Paper 6398.

Raymo, James M.; Mencarini, Letizia; Iwasawa, Miho and Rie Moriizumi (2010): "Intergenerational Proximity and the Fertility Intentions of Married Women," *Asian Population Studies*, 6(2), pp 193-214.

Rupert, Peter; Stancanelli, Elena and Etienne Wasmer (2009): "Commuting, Wages, and Bargaining Power," Annales d'Economie et de Statistiques, (95-96), pp 201-221.

SOEP (2005): "DTC- Desktop Companion to the German Socio-Economic Panel (SOEP)," edited by John P. Haisken-DeNew (RWI Essen) & Joachim R. Frick (DIW Berlin), Version 8.0 - Dec 2005 Updated to Wave 21 (U).

Stern, Steven (1995): "Measuring Child Work and Residence Adjustments to Parents' Long-Term Care Needs," *The Gerontologist*, 36(1), pp 76-87.

Statistische Ämter des Bundes und der Länder (2011): "Kindertagesbetreuung regional 2011, Vergleich aller 412 Kreise in Deutschland," Statistisches Bundesamt, Wiesbaden.

UNESCO (1997): International Standard Classification of Education - ISCED 1997, http : //www.unesco.org/education/information/nfsunesco/doc/isced₁997.htm, November 1997.

US Census Bureau (2011): "Child Care Costs in the Current Population Survey's Annual Social and Economic Supplement (CPS ASEC): A Comparison to SIPP," SEHSD Working Paper Number 2011-1, Suzanne Macartney and Lynda Laughlin, Housing and Household Economic Statistics Division, U.S. Census Bureau.

Wagner, Gert G., Joachim R. Frick, and Jürgen Schupp (2007): "The German Socio-Economic Panel Study (SOEP) - Scope, Evolution and Enhancements," *Schmollers Jahrbuch* 127(1), pp.139-169.

Walker, James R. (1991): "Public Policy and the Supply of Child Care Services," in *The Economics of Child Care* edited by David M. Blau, Russell Sage Foundation, New York.

Zamarro, Gemma (2011): "Family Labor Participation and Child Care Decisions. The Role of Grannies," RAND Working Paper WR-833.

A Appendix



Figure A-1: Grandparent-provided Care for grandchildren, ≤ 6



Figure A-2: Frequency of care for and distance to grandchild, ≤ 6





Figure A-3: Frequency of care for and distance to grandchild, ≤ 6

Data: Survey of Health, Ageing and Retirement in Europe (SHARE), 2^{nd} wave.

Figure A-4: Frequency of care for and distance to grandchild, ≤ 6



Data: Survey of Health, Ageing and Retirement in Europe (SHARE), 2^{nd} wave.

	Women	Mothers
	25-50	25-50
Age	37.4 (7.1)	38.8 (6.6)
30-34	0.20	0.18
25.20	0.20	0.24
40.44	0.22	0.24
45 50	0.21	0.23
Married living together	0.68	0.23
Other than Corman nationality	0.03	0.01
Children	0.76	1
	0.10	0.12
Children 2.6	0.14	0.13
N = h = f Chille	0.14	0.19
Number of Children	1.5(1.2)	2.0(0.9)
Primary education	0.01	0.01
Tertiary education	0.27	0.26
Regular fulltime job	0.36	0.25
Regular part time job	0.24	0.29
Small community	0.45	0.48
Large community	0.29	0.26
in East Germany	0.25	0.28
Parents or in-laws in same house	0.14	0.13
Parents or in-laws close	0.42	0.44
- Parents or in-laws in same neighborhood	0.19	0.21
- Parents or in-laws in same town	0.23	0.24
Parents or in-laws far away	0.44	0.43
- Parents or in-laws one hour away	0.29	0.29
- Parents or in-laws further away	0.15	0.13
Parents or in-laws in foreign country	0.01	0.01
Children in non-relative care (nursery, sitter, paid care etc.)***	-	0.65
- Children (< 3)	-	0.14
- Children (3 – 6)	-	0.84
Children cared for by relatives***	-	0.34
- Children (> 3)	-	0.33
- Children (3 - 6)	-	0.34
Spouse's income*	2844.5 (2362.0)	2865.0 (2159.0)
Hourly wage**	12.4 (6.2)	12.1 (6.2)
Tenure in firm**	8.4 (7.3)	8.9 (7.6)
Commuter****	0.50	0.49
N	10,732	8,129

Table A.1: Means (Std.) - SOEP pooled sample-- 1991, 1996, 2001, 2006-

*** Only available for 2001 and 2006 (N = 761 for < 3, N1296 for 3 - 6.) *Only taking into account strictly positive income (N = 7, 323, N = 6, 083 for women, mothers) **Among those working regular part-or fulltime jobs (N = 6, 471, N = 4, 348)**** Only available for 2001 and 2006 (N = 4363, N = 2913).

=

	Mothers with	Working Mothers with	Mothers with	Working Mothers with
	children $\leq = 6$	children $\leq = 6$	children < 3	children < 3
All	$34.9\% \ (n=2148)$	$44.4\% \ (n = 786)$	$33.3\% \ (n = 901)$	$46.6\% \ (n = 176)$
Parents or in-laws in same house	$49.8\% \ (n = 265)$	$64.6\% \ (n = 99)$	$48.6\% \ (n = 111)$	77.8% $(n = 18)$
Parents or in-laws close	$45.1\% \ (n = 990)$	$53.8\% \ (n = 379)$	$43.7\% \ (n = 398)$	57.6% $(n = 85)$
- Parents or in-laws in same neighborhood	$50.7\% \ (n = 489)$	56.8% $(n = 190)$	$49.5\% \ (n=204)$	$60.5\% \ (n = 43)$
- Parents or in-laws in same town	$39.5\% \ (n = 501)$	$50.8\% \ (n = 189)$	$37.6\% \ (n = 194)$	54.8% $(n = 42)$
Parents or in-laws far away	$19.1\% \ (n = 893)$	$26.3\% \ (n = 308)$	$18.4\% \ (n = 392)$	$26.0\% \ (n = 73)$
- Parents or in-laws one hour away	24.7% ($n = 635$)	$32.1\% \ (n=224)$	$24.6\% \ (n = 280)$	$32.7\% \ (n = 52)$
- Parents or in-laws further away	$5.4\% \ (n = 241)$	$11.3\% \ (n = 80)$	$2.8\% \ (n = 107)$	10.0%(n=20)
Parents or in-laws in foreign country	$5.9\% \ (n = 17)$	0 (n = 4)	$0 \ (n = 5)$	$0 \ (n = 1)$

Table A.2: Use of relative-provided child care by proximity to parents or in-laws (number of observations)

Women 25-50, pooled waves 2001 and 2006.

Table A.3: Use of relative-provided child care, age, health status and labor forceparticipation of grandmothers (number of observations)

	All		Close to own parent	ts and using relative-provided care
	Mothers with	Working Mothers with	Mothers with	Working Mothers with
	children $\leq = 6$	children $\leq = 6$	children $\leq = 6$	children $\leq = 6$
Age grandmother	58.0 (7.8) $(n = 503)$	$57.9 \ (7.7)(n = 151)$	57.3 $(7.7)(n = 81)$	$55.4(7.4) \ (n = 33)$
Grandmother not employed	51.5% $(n = 410)$	$45.2\% \ (n = 126)$	$64.2\% \ (n=67)$	$61.5\% \ (n = 26)$
Grandmother works full time	$25.8\% \ (n = 410)$	$32.5\% \ (n = 126)$	$17.9\% \ (n = 67)$	$30.8\% \ (n = 26)$
Grandmother works part time	$18.0\% \ (n = 410)$	$15.9\% \ (n = 126)$	$13.4\% \ (n = 67)$	$7.7\% \ (n = 26)$
Grandmother very good or good health	30.2% (n = 410)	$21.4\% \ (n = 126)$	$28.4\% \ (n = 67)$	$26.9\% \ (n = 26)$
Grandmother not good or bad health	23.7%(n = 410)	23.0%(n = 126)	$28.4\% \ (n = 67)$	$26.9\% \ (n = 26)$

.

Women 25-50, pooled waves 2001 and 2006.

Table A.4: Marginal effects from Probit estimation for having children

Married, living together	0.286***	(0.012)
Other than German nationality	0.008	(0.027)
Log (Spouse's income) [†]	0.004***	(0.001)
in East Germany	0.177***	(0.023)
Tertiary education (ISCED: 5,6)	-0.089***	(0.010)
Parents or in-laws close	0.041***	(0.009)
Parents or in-laws in same house	0.002	(0.013)
Small community	0.036***	(0.011)
Large community	-0.065***	(0.012)
Observations	10,732	
\dagger Missing values and values < 1 are	set to 0. Standard error	s in parentheses:
*** p<0.01,** p<0.05,* p<0.1 Dat	a: SOEP unbalanced par	nel 91,96,01,06;

*** p<0.01,** p<0.05,* p<0.1 Data: SOEP unbalanced panel 91,96,01,06; Women 25-50. Regression includes year, state, and age group dummies. Reference group: unmarried women age 25-29, with education level 1 or 2(ISCED:0-4), living in West Germany in North Rhine-Westphalia in 1991, in a medium-sized town, far from parents or in-laws.

Table A.5: Marginal effects from Probit estimation for having children with interaction terms

Married, living together	0.286^{***}	(0.012)	0.286***	(0.012)
Other than German nationality	0.008	(0.027)	0.008	(0.027)
Log (Spouse's income) [†]	0.004^{***}	(0.001)	0.004***	(0.001)
in East Germany	0.178^{***}	(0.023)	0.178***	(0.023)
Tertiary education (ISCED: 5,6)	-0.102^{***}	(0.013)		
Primary/Secondary edu (ISCED: 0-4)			0.102***	(0.013)
Parents or in-laws close	0.032^{***}	(0.010)	0.064***	(0.016)
Parents in-laws close*Tert edu	0.031*	(0.017)		
Parents in-laws close*Non-Tert edu			-0.033*	(0.020)
Parents or in-laws in same house	0.001	(0.013)	0.001	(0.013)
Small community	0.036^{***}	(0.011)	0.036***	(0.011)
Large community	-0.065***	(0.012)	-0.065***	(0.012)
Observations	10,732		10,732	

 Observations
 10,732
 10,732

 †Missing values and values <1 are set to 0. Standard errors in parentheses: *** p<0.01, ** p<0.05,* p<0.1 Data: SOEP</td>

 unbalanced panel 91,96,01,06; Women 25-50. Regressions include year, state, and age group dummies. Reference group:

 unmarried women age 25-29 with education level 3 (ISCED 5,6) and education level 1 or 2(ISCED.04) respectively

 living in West Germany in North Rhine-Westphalia in 1991, in a medium-sized town, far from parents or in-laws.

Table A.6: Marginal effects from Probit estimation for mothers' labor force participation

	Regular Pa Fulltime Is	rt or
Number of children	-0.114***	(0.007)
Children 0-2	-0.398***	(0.016)
Married, living together	-0.128***	(0.017)
Other than German nationality	-0.123**	(0.054)
Log (Spouse's income) [†]	0.006^{***}	(0.002)
in East Germany	0.116^{**}	(0.046)
Tertiary education (ISCED: 5,6)	0.171^{***}	(0.014)
Parents or in-laws close	0.029**	(0.013)
Parents or in-laws in same house	0.022	(0.019)
Small community	-0.003	(0.015)
Large community	0.018	(0.017)
Observations	8,129	

 $_{6,123}^{+123}$ (Missing values and values < 1 are set to 0. Standard errors in parentheses;*** p<0.01, ** p<0.05, * p<0.1; Data: SOEP unbalanced panel, 91,96,01,06; Mothers 25-50. Reference group: unmarried mothers age 25-29 with education level 1 or 2(ISCED:0-4) in 1991, in a medium-sized town in West Germany North Rhine-Westphalia, far from parents or in-laws, with children age 3 or older. Regression includes year, state, and age group dummies.

Table A.7: Marginal effects from Probit estimation for labor force participation of (1) men and (2) single childless women

	Regular Pa	rt or	Regular Pa	rt or	
	Fulltime Jo	ь	Fulltime Jo	ъ	
	(1)		(2)		
Married, living together	0.106^{***}	(0.008)			
Other than German nationality	-0.096***	(0.032)	-0.097	(0.071)	
in East Germany	-0.067**	(0.027)	-0.299 * *	(0.136)	
Tertiary education (ISCED: 5,6)	0.088^{***}	(0.006)	0.135^{***}	(0.024)	
Parents or in-laws close	0.009	(0.007)	0.033	(0.028)	
Parents or in-laws in same house	-0.008	(0.010)	-0.100^{***}	(0.037)	
Small community	0.003	(0.009)	-0.034	(0.036)	
Large community	-0.013	(0.010)	-0.125^{***}	(0.033)	
Observations	8,653		1,176		
1) Men 25-50. Reference group: un	imarried men	age 25-29	with education	on level 1 or 2 (ISCED: 0-4) in 1991, in a	-

nedium-sized town in West Germany in North Rhine-Westphalia. Regressions include year, state, and age group dummies.

Table A.8: Coefficients of Heckman Selection Model for mothers' log monthly wages

	Log month wage	ly	Selection Equation		
	(1)		(2)		
Married, living together	-0.035**	(0.017)	-0.326***	(0.044)	
Number of children	-0.047^{***}	(0.010)	-0.287^{***}	(0.017)	
Other than German nationality	-0.116	(0.083)	-0.311**	(0.140)	
Tertiary education (ISCED: 5,6)	0.282^{***}	(0.016)	0.441^{***}	(0.036)	
Parents or in-laws close	-0.056^{***}	(0.015)	0.073^{**}	(0.033)	
Parents or in-laws in same house	-0.059***	(0.022)	0.056	(0.048)	
Small community	-0.038**	(0.018)	-0.007	(0.039)	
Large community	0.053^{***}	(0.020)	0.046	(0.044)	
in East Germany	-0.081*	(0.042)	0.296^{***}	(0.119)	
Log (Spouse's income) [†]			0.016^{***}	(0.005)	
Children 0-2			-1.105^{***}	(0.056)	
Tenure in firm	0.017^{***}	(0.001)			
Monthly hours worked	0.007^{***}	(0.000)			
Constant	5.787***	(0.055)	0.186^{**}	(0.087)	
Observations	8,129		8,129		

¹Missing values and values < 1 are set to 0. Standard errors in parentheses ^{***} p<0.01, ^{**} p<0.05, ^{*} p<0.1. Data: SOEP unbalanced panel 91,96,01,06; mothers 25-50. Reference group: unmarried mothers of age 25-29 of children age 3 or older with education level 1 or 2 (ISCED: 0-4) in 1991, in a medium-sized West German town in North Rhine-Westphalia , far from parents or in-laws. Regressions include age group, state, and year dummies.

Table A.9: Proximity to grandparents and hourly wages Coefficients of OLS estimation of mothers' log hourly wages

Married, living together	0.000	(0.015)
Number of children	-0.026***	(0.008)
Other than German nationality	-0.121	(0.077)
Tertiary education (ISCED: 5,6)	0.280***	(0.014)
Tenure in firm	0.016***	(0.001)
Parents or in-laws close	-0.052***	(0.014)
Parents or in-laws in same house	-0.052**	(0.020)
Small community	-0.040**	(0.017)
Large community	0.045**	(0.019)
in East Germany	-0.081*	(0.044)
Constant	1.955***	(0.041)
		. ,
Observations	4,348	
B-squared	0.262	

R-squared 0.262 Standard errors in parentheses: *** p<0.01, ** p<0.05, * p<0.1 OLS Estimation; Data: SOEP unbalanced panel 91,96,01,06; mothers 25-50 with full-or part time regular job. Regression includes age group, state, and year dummies. Reference group: unmarried mothers age 25-29 living in West Germany in North Rhine-Westphalia , with education level 1 or 2 (ISCED: 0-4), in 1991, in a medium-sized town, far from parents or in-laws.

Table A.10: Proximity to grandparents and hourly wages Coefficients of Heckman Selection Model for mothers' log hourly wages without variables posing a possible endogeneity problem: marital status and spouse's income

	Log hourly wage		Selection Equation		
	(1)		(2)		
Number of children	-0.038***	(0.010)	-0.292***	(0.017)	
Other than German nationality	-0.135*	(0.077)	-0.340**	(0.139)	
Tertiary education (ISCED: 5,6)	0.293^{***}	(0.015)	0.433^{***}	(0.036)	
Parents or in-laws close	-0.049^{***}	(0.014)	0.061*	(0.033)	
Parents or in-laws in same house	-0.051**	(0.020)	0.037	(0.047)	
Small community	-0.040**	(0.016)	-0.014	(0.038)	
Large community	0.046^{**}	(0.019)	0.056	(0.044)	
in East Germany	-0.067	(0.044)	0.314^{***}	(0.119)	
Children 0-2		. ,	-1.117^{***}	(0.056)	
Tenure in firm	0.016^{***}	(0.001)			
Constant	1.897***	(0.047)	0.061	(0.083)	
Observations	8,129		8,129		

Standard errors in parentheses *** p < 0.01, ** p < 0.05, * p < 0.1. Heckman Selection Model; Data: SOEP unbalanced panel 91,96,01,06; mothers 25-50. Reference group: mothers of age 25-29 of children age 3 and older living in West Germany, with education level 1 or 2 (ISCED: 0-4) in 1991, in a medium-sized West German town in North Rhine-Westphalia , far from parents or in-laws. Regressions include age group, state, and year dummies.

Table A.11: Proximity to grandparents and hourly wages Coefficients of Heckman Selection Model for low educated mothers' log hourly wages

	Log hourly		Selection	
	(1)		Equation	
	(1)	(0.010)	(2)	(0.071)
Married, living together	-0.002	(0.019)	-0.341***	(0.051)
Other than German nationality	-0.207**	(0.090)	-0.435^{***}	(0.155)
Number of children	-0.038***	(0.012)	-0.290***	(0.020)
Parents or in-laws close	-0.042**	(0.017)	0.059	(0.038)
Parents or in-laws in same house	-0.038	(0.024)	0.035	(0.054)
Small community	-0.043**	(0.020)	-0.013	(0.044)
Large community	0.042^{*}	(0.022)	0.048	(0.050)
in East Germany	-0.135^{***}	(0.048)	0.477^{***}	(0.117)
Children 0-2			-1.102^{***}	(0.070)
Log (Spouse's income) [†]			0.018^{***}	(0.006)
Tenure in firm	0.017^{***}	(0.001)		
Constant	1.895***	(0.055)	0.190*	(0.097)
Observations	6,035		6,035	

 $\begin{array}{c} 0 \text{b}_{0,035} & \text{b}_{0,035} \\ \hline \text{Standard errors in parentheses} & *** p < 0.01, ** p < 0.05, * p < 0.1. Heckman Selection Model; Data: SOEP unbalanced panel 91,96,01,06; mothers 25-50 with primary pr secondary education. Reference group: mothers of age 25-29 of children age 3 and older living in West Germany, with education level 1 or 2 (ISCED: 0-4) in 1991, in a medium-sized West German town in North Rhine-Westphalia , far from parents or in-laws. Regressions include dummies for age groups, states and years. \\ \end{array}$

Table A.12: Proximity to grandparents and hourly wages Coefficients of Heckman Selection Model for high educated mothers' log hourly wages

	Log hourly		Selection	
	wage		Equation	
	(1)		(2)	
Married, living together	-0.015	(0.028)	-0.286^{***}	(0.095)
Number of children	-0.026	(0.016)	-0.265^{***}	(0.038)
Other than German nationality	0.089	(0.148)	0.220	(0.353)
Parents or in-laws close	-0.066***	(0.024)	0.090	(0.068)
Parents or in-laws in same house	-0.069*	(0.038)	0.159	(0.107)
Small community	-0.038	(0.030)	-0.029	(0.084)
Large community	0.047	(0.033)	0.042	(0.091)
in East Germany	-0.146^{**}	(0.064)	0.307*	(0.174)
Children 0-2			-1.095^{***}	(0.098)
Log (Spouse's income) [†]			0.007	(0.011)
Tenure in firm	0.013^{***}	(0.002)		
Constant	2.266***	(0.088)	0.679^{***}	(0.208)
Observations	2,094		2,094	

 Observations
 2,094
 2,094

 Standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1 Heckman Selection Model; Data: SOEP unbalanced panel</td>

 91,96,01,06; mothers 25-50, with tertiary education. Reference group: mothers of age 25-29 with children age 3 and older living

 in West Germany, with education level 3 (ISCED: 5-6) in 1991, in a medium-sized West German town in North Rhine-Westphalia, far from parents or in-laws. Regressions include age group, state, and year dummies.

Table A.13: Grandparent-provided child care and commutes Marginal effects from Probit Estimation for mothers without variable posing a possible endogeneity problem: Children in non-relative care

	Commuter	
Married, living together	0.006	(0.055)
Number of children	0.053	(0.033)
Other than German nationality	-0.278**	(0.127)
Children 0-2	-0.024	(0.060)
Log (Spouse's income) [†]	-0.003	(0.011)
Tertiary education (ISCED: 5,6)	0.126**	(0.051)
Tenure in firm	0.004	(0.004)
Children cared for by relatives	0.088*	(0.048)
Small community	0.263***	(0.055)
Large community	-0.213***	(0.065)
in East Germany	-0.228*	(0.135)
Observations	537	

Table A.14: Grandparent-provided child care and commutes Marginal effects from Probit Estimation for (1) low educated and (2) high educated mothers

	Commuter		Commuter	
	(1)		(2)	
Married, living together	0.003	(0.071)	0.015	(0.095)
Number of children	0.057	(0.042)	0.022	(0.056)
Other than German nationality	-0.438^{***}	(0.108)	-0.016	(0.217)
Children 0-2	-0.148*	(0.086)	0.125	(0.093)
Log (Spouse's income) [†]	-0.019	(0.015)	0.019	(0.018)
Tenure in firm	0.008	(0.005)	-0.000	(0.008)
Children cared for by relatives	0.097	(0.061)	0.106	(0.088)
Children in non-relative care	0.017	(0.086)	-0.098	(0.133)
Small community	0.268^{***}	(0.068)	0.260 * *	(0.106)
Large community	-0.222^{***}	(0.082)	-0.264 * *	(0.119)
in East Germany	-0.280*	(0.156)	0.125	(0.232)
Observations	352		183	